Author's response to reviews

Title: Evaluation of the effect of hydroalcoholic leaves extract of indigofera spicata forssk. (fabaceae) on blood glucose level of normal, oral glucose loaded and alloxan induced diabetic rodents

Authors:

Eshetie M Birru (meshetie21@gmail.com)
Mohammedbrhan A Ata (mohammedB18@gmail.com)
Zewdneh S Sabe (zeedshow@gmail.com)

Version: 2 Date: 21 April 2015

Author's response to reviews:

@. The last paragraph of the background portion had minor modified
@. In some segments of the main body of the manuscript which were not double lined in spacing are corrected
@. competing interest is incorporated between the conclusion and authors contribution